HONG KONG UNIVERSITY OF SCIENCE & TECHNOLOGY

COMP 102 : Pr ogramming & Computer Fundamentals I

Summer 1996

Laboratory Assignment 5

29 & 31 July

Write a pr ogram that allows two people to play the game "tic-tac-toe". It is a game in which two players alternately put Xs and Os in a 3x3 grid. Use a 2D array to r epresent the grid. Y our pr ogram should be able to detect the winner if either: • 3 Xs (or 3 Os) ar e in a r ow, • 3 Xs (or 3 Os) ar e in a column, or • 3 Xs (or 3 Os) ar e in a diagonal.

A typical output of this pr ogram will be:

A B C 1 O . X 2 . O . 3 . . X

O's turn: 3 B

A B C 1 O . X 2 . O . 3 . O X

X's turn: 2 C

A B C 1 O . X 2 . O X 3 . . X

Player X won.